Author's response to reviews

Title: Parasite threshold associated with clinical malaria in areas of different transmission intensities in north eastern Tanzania

Authors:

Bruno P Mmbando (bmmbando@tanga.mimcom.net)
John P Lusingu (jpalusingu@yahoo.co.uk)
Lasse S Vestergaard (lvcmp@rh.dk)
Martha M Lemnge (mlemnge@tanga.mimcom.net)
Thor G Theander (theander@cmp.dk)
Thomas H Scheike (t.sheike@biostat.ku.dk)

Version: 5 Date: 28 October 2009

Author's response to reviews: see over
Dear Editor,

Re: MS 2021230647268477.
Attached is a revised manuscript entitled - Parasite threshold associated with clinical malaria in areas of different transmission intensities in north eastern Tanzania.

We have gone through the manuscript done the corrections on reference no 21, 24 and 25 as requested. The proofreading of the entire paper was also done and we now fill that it can be published as it stands.

We have also attached a bibliography generated by the latex.

On behalf of all authors,

Bruno P. Mmbando

email: bmmbando@tanga.mimcom.net